From: Cherry, Brian K

Sent: 1/17/2012 8:07:16 AM

To: 'Clanon, Paul' (paul.clanon@cpuc.ca.gov)

Cc:

Subject: Re: L-1101 outage- OIS 495538- .L-1101m Final update

Yes. Looks like it.

From: Clanon, Paul [mailto:paul.clanon@cpuc.ca.gov]

Sent: Tuesday, January 17, 2012 8:02 AM

To: Cherry, Brian K

Subject: Re: L-1101 outage- OIS 495538- .L-1101m Final update

Getting uncomfortably close to my house. So this is two separate incidents last night, both involving vaults?

On Jan 17, 2012, at 7:58 AM, "Cherry, Brian K" < BKC7@pge.com > wrote:

From: Redacted

Sent: Monday, January 16, 2012 11:55 PM

To: SF Update

Subject: RE: L-1101 outage- OIS 495538- .L-1101m Final update

At 2330 hrs all customers restored on the L-1101 circuit. Cable to be repaired at a later date.

Redacted

Bay Area Restore & Control Lead Supervisor

Distribution System Operations
Cell: Redacted
Dodastod
From: Redacted Sent: Monday, January 16, 2012 11:31 PM
To: SF_Update Subject: RE: L-1101 outage- OIS 495538-
<b>Subject.</b> RE. L-1101 outage- 013 493336
Undete on L 1101 outons. Cohlemen found foiled poble in Man hale 6426. All
Update on L-1101 outage- Cablemen found failed cable in Man hole 6436. All customers will be picked up within 30 to 45 minutes. Area served is the 2200
block of Lyon, 3700 block of 19th street.,
Redacted
Bay Area Restore & Control Lead Supervisor
Distribution System Operations
cell: Redacted
cen.

From: Redacted

**Sent:** Monday, January 16, 2012 11:05 PM

To: SF\_Update Subject: RE: L-1101 outage- OIS 495538- . 89 Customers will remain out until Cable men can get in the hole for further review North East corner of 19th and Sloat. From: Redacted Sent: Monday, January 16, 2012 10:42 PM To: SF\_Update Subject: RE: L-1101 outage- OIS 495538- wrong date stamped on outage. The L-1101 outage has been reduced to 89 customers. From: Redacted Sent: Monday, January 16, 2012 10:35 PM To: SF\_Update Subject: RE: L-1101 outage- OIS 495538- wrong date stamped on outage. Another 3787 restored at 10:25. Gov Rel has notified CCSF OES and will continue to provide updates as needed..

Redacted

Bay Area Restore & Control Lead Supervisor

Distribution System Operations

cell: Redacted
From: Redacted  Sent: Monday, January 16, 2012 10:17 PM  To: SF_Update  Subject: RE: L-1101 outage- OIS 495538- wrong date stamped on outage.
879 customers have been picked up. System Operators working with T-Men , Cablemen and Supervisors in the field to sectionalize and restoring remaining 4800+ customers.
From: Redacted Sent: Monday, January 16, 2012 9:49 PM To: SF_Update Cc Redacted Subject: RE: L-1101 outage- OIS 495538- wrong date stamped on outage.
Man Hole smoking North East corner of 19 <sup>th</sup> and Sloat. San Francisco Fire Department on site.

From: Redacted

To: SF\_Update
Cc: Redacted

Sent: Monday, January 16, 2012 9:45 PM

Subject: RE: L-1101 outage- OIS 495538- wrong date stamped on outage.

SB\_GT&S\_0218833

OIS is indicating an First no light of 1-14-12. Incorrect information. Requesting Tech Team to correct the date as is should read a date of 1-16-12. System Operators are switching on troubled circuit.
Redacted  Bay Area Restore & Control Lead Supervisor
Distribution System Operations
cell: Redacted
From: Redacted Sent: Monday, January 16, 2012 9:29 PM To: SF_Update Subject: L-1101 outage- OIS 495538
All:
Experiencing an outage on the L-1101 circuits. First no light, 2116. No Hazards reported at this time. Two Troublemen, 1 Cableman and I Restoration Supervisor in Route. General Vicinity of Portal Avenue, Maywood Dr, Crestlake, Wawonna aves.
Redacted
Bay Area Restore & Control Lead Supervisor
Distribution System Operations
Redacted